Author's response to reviews

Title: Privacy-preserving record linkage using Bloom filters

Authors:

Rainer Schnell (Rainer.Schnell@uni-due.de)
Tobias Bachteler (Tobias.Bachteler@uni-due.de)
Jörg Reiher (Joerg.Reiher@uni-due.de)

Version: 5 Date: 17 August 2009

Author's response to reviews:

In the section "Results for actual databases" we corrected the figure numbers and added a reference sentence to figure 7: (figure 7 shows a cutout of figure 6 to highlight recall levels above .75)